Author's response to reviews

Title: Cystic Fibrosis and Renal Disease

Authors:

  Baha A Al-Shawwa (balshaww@mcw.edu)
  Aparna R Rao (arrao@mcw.edu)

Version: 3  Date: 16 May 2007

Author's response to reviews: see over
Date: May 16, 2007

To: The Editorial Team
     Journal of Medical Case Reports

**Reference: Resubmission of manuscript # 9166599801393571. “Cystic Fibrosis and Renal Disease”**

To,
The editorial team of Journal of Medical Case Reports,

These are the responses to the reviewer comment and changes have been made in the manuscript to reflect that:

Point 1: “Is” was added as recommended in page 2 line 1.

Point 2: Phrase “by a random urine sample” was added to page 3, last paragraph. Also a phrase “and no corticosteroids were used” was added to the 1st line of page 4.

Point 3 and 4: Urine and kidney biopsy results were added as the reviewer suggestion in the second paragraph on page 4. In reference to the medication that the patient was receiving, a sentence was added in page 3, 3rd paragraph.

Point 5: the sentence in page 5, line 1 was rearranged as recommended.

We thank the reviewer for these suggestions. We think that the revised manuscript is significantly improved and hopefully the Journal of Medical Case Reports will consider it acceptable for publication.

Sincerely,

Baha A. Al-Shawwa, MD
Corresponding author

Aparna R. Rao, MD